

Level and correlates of disrespect and abuse among newborns in selected public hospitals



Rediet Gezahegn^{1*} and Abiy Seifu Estifanos¹

of Addis Ababa, Ethiopia

Abstract

Background The provision of respectful and dignified maternal and newborn care is an important component of the quality of childbirth care. Although a growing body of evidence was generated on disrespect and abuse (D&A) of women during childbirth in the past decade there is limited evidence on D&A experienced by newborns. Our study aimed to determine the level of and factors associated with D&A among newborns.

Methods We conducted the study in three public hospitals in Addis Ababa. We directly observed childbirth care starting from the first stage of labor through two hours after the birth of 498 mother–baby dyads. We used frequencies and percentages to describe different forms of D&A among newborns. We used binary and multivariable logistic regression analysis to assess the association between the D&A among newborns and independent variables.

Result All of the newborns 496/496 (100%) experienced at least one form of D&A. Physical abuse was experienced by 41.1% of newborns in the form of unnecessary airway suctioning (23.2%) or slapping or holding upside down (33.5%). Additionally, 42.3% weren't dried immediately after birth, 9.1% weren't placed on the mother's abdomen skin-to-skin, 61.7% had their cord cut before 1 min of birth, 34.9% weren't breastfed within an hour of birth, 24.2% didn't receive vitamin K and 1.8% didn't receive tetracycline. All newborns who developed complications (69/69) received treatments without the consent of parents/caregivers. Moreover, 93.6% of parents/caregivers didn't receive explanations regarding newborn care while the lack of breastfeeding counseling and thermal support during the immediate post-partum period was 87.3%. The likelihood of D&A was higher among newborns who were preterm (AOR = 2.02; 95% CI: 1.11–3.69), female (AOR = 2.01; 95% CI: 1.37–2.95), delivered assisted by instrument (AOR = 2.19; 95%CI: 1.20–3.99), whose mothers reside in rural areas (AOR = 1.97; 95%CI: 1.22–3.20), born from unmarried mothers (AOR = 2.77; 95%CI (1.26–6.06) and whose mothers received fewer than four-time antenatal care (ANC) visits (AOR = 2.37; 95%CI: 1.42–3.96).

Conclusion Our study found a high magnitude D&A among newborns. Gestational age at birth, sex of the newborn, maternal residence, maternal marital status, number of ANC visits, and mode of delivery were statistically significantly associated with D&A among newborns.

Keywords Disrespect and abuse, Respectful newborn care, Newborn care

*Correspondence: Rediet Gezahegn redinagezahegn@gmail.com Full list of author information is available at the end of the article



© The Author(s) 2023. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.gr/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.gr/licenses/by/4.0/. The Creative Commons Public Domain and credit line to the data.

Plain English Summary

The provision of respectful and dignified care to mothers and newborns during antenatal, childbirth, and postnatal periods is crucial to improve their health outcomes and experiences. Although there is a growing body of knowledge on respectful maternity care, there is limited evidence that systematically assessed newborns' experiences of care.

We measured the level of disrespect and abuse among newborns and assessed factors that increase the likelihood of experiencing disrespect and abuse among newborns born in three hospitals in Addis Ababa. We directly observed the care provision to assess whether the newborn was (1) physically abused (suction of the nose or mouth without medical indication; shaken, slapped, or held upside down), (2) stigmatized or discriminated (care compromised because the newborn was considered too sick to save, HIV exposed, or has congenital malformation), (3) didn't receive standard care, and (4) mothers/caregivers didn't receive counseling on breastfeeding and thermal care.

We found that all of the newborns have experienced at least one form of disrespect and abuse during childbirth or the postpartum period. 41% experienced physical abuse; 99.6% didn't receive standard care; 87.3% didn't receive counseling on breastfeeding and thermal care; and 13%, 75%, and 67% of the newborns who were identified as too sick to save, HIV exposed, and congenital malformation, respectively, were ignored, denied care or treated differently. In addition, the likelihood of experiencing disrespect and abuse was high among newborns who were preterm, female, from rural areas, of unmarried mothers, whose mothers received fewer than four ANC visits and delivered by instrument.

Background

The World Health Organization (WHO) framework for the quality of maternal and newborn care consists of two interlinked dimensions; the quality of provision of care and the quality of experience of care by women, newborns, and their families. The framework also identified eight domains including the provision of evidence-based care, effective communication, respect and dignity, and emotional support [1]. The provision of respectful care for newborns is increasingly considered as an important component of providing dignified care. In 2019, the White Ribbon Alliance incorporated newborn babies in the updated respectful maternity care charter by recognizing that women and newborns have independent and inalienable basic human rights that must be respected and guaranteed [2].

Ethiopia is among the world's top countries with the highest burden of neonatal mortality rate [3]. Between 2016 and 2019 the proportion of childbirths assisted by skilled providers in Ethiopia increased from 28 to 50%. However, during the same period neonatal mortality showed an increase from 29 to 33 per 1000 live births [4, 5]. This implies that unless the quality gap is closed through the provision of effective care for all women and newborns delivering in the facilities [6, 7] an increase in the proportion of births assisted by skilled attendants may not translate into saving lives [8]. Hence, the provision of high-quality care for women and newborns that addresses their rights, needs, experiences, and preferences is critical to attaining the national and Sustainable Development Goals (SDG) targets [7].

Women's perception or experience of disrespect and abuse (D&A) during childbirth in the facility influences their choice about where to deliver and could prevent them from seeking health care services timely or at all increasing the risk of poor childbirth outcomes to themselves and their newborns [9-11]. Even though newborns can't communicate verbally and give consent about their needs and preferences for tests, treatments, and referrals, their parents or caregivers have the right to receive information and provide informed consent for their child's care in the child's best interests. To address this the healthcare system should ensure the health, safety, and dignity of both the women and their newborns [2, 12].

D&A among newborns was defined only recently. In 2017, using the seven categories developed for the mistreatment of women during childbirth, a comprehensive systematic review was done to define D&A among newborns that identified eight categories. These are physical abuse, verbal abuse, stigma/ discrimination, failure to meet professional standards of care, the poor rapport between patients and providers, health system conditions and constraints, legal accountability, and bereavement and posthumous care [13]. The systematic review offered a framework for assessing disrespect, abuse, and stigmatization among newborns.

Notwithstanding that, previous studies reported various forms of D&A among newborns including unnecessary suctioning of the airway [14, 15], slapping/holding the newborn upside down [15, 16], blaming of the mother for poor neonatal outcome [17],

discrimination of small or human immunodeficiency virus (HIV) exposed newborns [18, 19], separation of the mother and newborn without medical indication [15, 20], provision of substandard essential newborn care (ENC) [21–23] and abandonment of health care [24]. Very few studies measured the coverage of respectful newborn care [25] and the level of mistreatment among newborns [14, 15, 20]. However, evidence that comprehensively assessed the level of D&A among newborns using direct observation of childbirth care is lacking. Our study aimed to determine the level of and factors associated with D&A among newborns through a comprehensive assessment of D&A among newborns using direct observation of childbirth care.

Methods

Study design and setting

We conducted a facility-based cross-sectional study in three public hospitals in Addis Ababa. The data were collected from 21 July to 18 September 2020. Based on United Nations estimates Addis Ababa had an estimated 4,794,000 population in 2020 [26]. The 2019 Mini Ethiopian Demographic and Health Survey (EDHS) report showed that 97% of the pregnant women in Addis Ababa received ANC at least once, 82% received four and more ANC visits, 96% were attended by a skilled provider during delivery, and 74% of mothers and newborns received a postnatal check within the first 2 days after birth [5].

For our study, we selected three hospitals located in Addis Ababa namely Gandhi Memorial Hospital (GMH), Tikur Anbessa Specialized Hospital (TASH), and Ras Desta Memorial Hospital (RDMH). GMH provides childbirth care services in nine separate rooms; each room has a single bed for labor follow-up and attending delivery. Whereas, TASH and RDMH provide childbirth care services in shared rooms with six and five beds, respectively, in the labor wards and four and three delivery couches, respectively, in the delivery wards. In the three hospitals, maternal and newborn health (MNH) care services are provided mainly by midwives, final-year medical students (interns), and obstetrics/gynecology resident physicians.

Study participants

Study participants were mothers-baby dyads who delivered at the three hospitals during the data collection period. We included mothers who had live birth at the gestational age of 28 weeks and later and excluded mothers who had a stillbirth or live birth but whose neonates died within 2 h after delivery.

Sample size determination

We used a single population proportion formula to calculate the sample size required to estimate the magnitude of D&A among newborns. Since there is no previous estimate, we assumed that 50% of newborns experience at least one form of D&A, 95% confidence interval (CI), and 5% margin of error. By considering a 10% non-response rate, we estimated a sample size of 427. To measure the association of exposure variables with D&A among newborns, we used a double population formula using variables previously identified to have an association with D&A among mothers. We considered the exposure variable (monthly income) that gave us the largest sample size (n=544) using assumptions including 95% CI, 80% power, 10% non-response rate, and an adjusted odds of 1.74 (the odds of experiencing D&A among mothers with a monthly income of <2000 Ethiopian Birr (ETB) was 1.74 times compared to mothers who earn 2000 or more ETB) [27]. We took 544 as our final sample size.

Sampling procedure and data collection

We selected the three hospitals based on the volume of deliveries they manage to capture variation in the level and experience of D&A. Accordingly, we selected GMH, TASH, and RDMH from public hospitals in the city based on the high, medium, and low volume of deliveries they manage, respectively. Based on the reports from Health Management and Information System (HMIS), GMH manages 900-1200 deliveries monthly, TASH 500-600, and RDMH 200-300. The allocation of the sample to the hospitals was made proportional to their delivery volume. Thus 314, 150, and 80 participants were allocated for GMH, TASH, and RDMH, respectively. Given the limited resources we had, we employed a convenience sampling technique to recruit mothers. In a situation where more than one mother who fulfills the inclusion criteria were admitted to the labor ward, we selected one of the mothers randomly using a lottery method. When the observation for the selected mother was completed, we repeated the same process for the next mother.

We used a standard observation tool to directly observe forms of D&A among newborns. Based on Emma Sacks's systematic review definition of D&A among newborns; the observation tool included four categories (physical abuse, stigma/discrimination, failure to meet professional standard of care, and poor rapport between patients and providers). Each of the categories has verification items [13]. We excluded other domains for logistics reasons. To capture the provision of recommended evidencebased newborn health care services more broadly, we added to the observation tool additional items including drying immediately after birth, cord care, administration of vitamin K and tetracycline eye ointment, initiation of skin-to-skin contact and breastfeeding within an hour after birth from the WHO safe childbirth checklist [28] and ENC guidelines [29]. We used a structured questionnaire that was initially prepared in English and translated into Amharic to interview mothers to collect information about the socio-demographic and obstetric characteristics of mothers. We also extracted the health-related characteristics of newborns and obstetric characteristics of mothers from their medical records.

Six data collectors, four male and two female nurses with Bachelor of Science (BSc) degree training, who were not staff of the selected public hospitals collected the data through direct observation of care, interview of mothers, and extraction of data from the maternity registers and charts. Two data collectors were assigned to each of the three hospitals. RG supervised the data collection. The training was given to the data collectors for two days before data collection. The training covered the objective of the study, data collection ethics, and detailed contents of data collection tools and procedures. We conducted a pre-test of the tools and data collection procedure before the actual data collection. The observation was started during the first stage of labor and continued for two hours after birth. We informed the health facilities and MNH care providers about the objective of the study. However, to minimize observation bias, the data collectors didn't show them the items included in the observation tools. Although the data collectors were health workers with experience in childbirth care, their role was intentionally limited to observing the care and collecting data; they did not engage in the provision of care or giving feedback to the health care providers or mothers.

Data processing and analysis

The data were entered using EpiData V.4.0 software and exported to Stata V.14 software for cleaning and analysis. Frequencies, means, standard deviations, and percentages were calculated to describe socio-demographic and obstetric characteristics of mothers and health-related characteristics of newborns.

We coded the items for each domain of D&A as "0" if not practiced and "1" if practiced. To generate a composite variable for a specific domain of D&A; all of the items under the domain were added together. If a mother-newborn dyad experienced at least one of the items under the specific domain of D&A, that domain of D&A was considered as experienced.

For the multivariable logistic regression analysis; we performed factor analysis as an intermediate step to reduce the number of items used to assess D&A among newborns. We used a total of 19 items to assess D&A among newborns. In the initial factor analysis, we dropped six items. Three items were dropped because they had missed data for >10% of the observations and the other three items were dropped because of lack of variance across observations. The remaining 13 items were loaded across five factors with an Eigenvalue of >1. After Orthogonal rotation, Cronbach's alpha test was performed for the 13 items to measure the internal consistency which yielded a reliability coefficient of 0.6038. For each of the retained factors, Cronbach's alpha test was performed among items that had a factor loading of >0.4 or <-0.4. Except for Factor 1, which had a reliability coefficient of >0.7, the other factors were dropped due to low-reliability coefficients. We used Factor 1 as a measure of our outcome variable for the multivariable logistic regression analysis. Five items loaded more on Factor 1 which we labeled as "Physical abuse and substandard care" or simply as D&A among newborns. We dichotomized the outcome variable by adding the score for the five items. Those who didn't experience any of the items were categorized as 'No D&A' and those who did experience at least one of the items were categorized as 'D&A'. Table 1 presents the interpreted factor, items that had a factor loading of > 0.4, and the value of factor loadings.

We did binary and multivariable logistic regression analyses to assess the association between the

Table 1 Items of D&A among newborns which had a value of factor loading > 0.4

Items ^a	Items that loaded on Factor 1 "Physical abuse and substandard care"
Suctioning without medical indication	0.6169
Shaking or slapping or turning upside down the newborn	0.8093
The newborn is not dried immediately after birth and is not well covered with a clean and dry towel	0.5023
Vitamin K was not administered	0.8036
Crowded conditions, shared beds	0.7097

^a Items which had factor loading > 0.4

independent variables and the outcome variable. The examined independent variables include; socio-demographic characteristics of the woman (age, marital status, residence, occupation, educational status), obstetric characteristics of the woman (parity, number of ANC visits, mode of delivery, length of labor, any complication), and health-related characteristics of newborns (sex, gestational age at birth, birth weight, any complication).

In the multivariable regression models, we included independent variables with p-value of < 0.25 in the binary logistic regression and variables which had a statistically significant association with D&A from the literatures. We checked for multicollinearity among the independent variables using the variance inflation factor. We included all independent variables in the models since all of them had a variance inflation factor of fewer than 5.

Result

We invited a total of 544 mothers to participate in the study, and 498 consented to participate yielding a response rate of 498/544 (91.5%). During data cleaning, we discarded incomplete data from two participants. Thus, we analyzed the data from 496 mother-baby dyads.

Socio-demographic characteristics of mothers

The mean age of the mothers was 27.2 years (SD=4.5); 315 (63.5%) of mothers were 25–34 years old. The majority of mothers were married [440/496 (88.7%)], attended secondary school and above [289/496 (58.3%)], and were housewives [364/496 (73.4%)] (Table 2).

Obstetrics characteristics of mothers

Among the mothers we observed [312/496 (62.9%)] were first-time mothers. Almost all, [491/496 (99.0%)] had at least one ANC visit for their current pregnancy, [382/496 (77.8%)] had four or more ANC visits. The majority of mothers [413/496 (83.3%)] gave birth vaginally either spontaneously [279/496 (56.3%)] or assisted by forceps/vacuum [134/496, (27.0%)]. The remaining [83/496 (16.7%)] gave birth through Cesarean section. The mean stay at the hospital from admission until the end of the observation was 5.9 h (SD=3.4) (Table 3).

Health-related characteristics of newborns

More than one-third [341/496 (68.8%)] of the newborns were born term (37–41 weeks) whereas [84/496 (16.9%)] and [71/496 (14.3%)] of them were born preterm (28–<37 weeks) and post-term (>41 weeks), respectively. More than half, [259/496 (52.2%)] of the newborns were female, [94/496 (18.9%)] had low birth weight (\leq 2500gram) with a mean birth weight of 3,019gram (SD=507). Sixty-nine

Page 5 of 11	l
--------------	---

Table 2Socio-demographiccharacteristicsofmothersinselected public hospitals of Addis Ababa, Ethiopia, 2020 (n = 496)

Variables	Frequency	Percent
Age		
<25	148	29.8
25-34	315	63.5
≥35	33	6.7
Place of residence		
Urban	380	76.6
Rural	116	23.4
Religion		
Orthodox	211	42.5
Muslim	186	37.5
Protestant	91	18.4
Catholic	2	0.4
Traditional	6	1.2
Marital status		
Unmarried	28	5.7
Married/living together	440	88.7
Divorced	22	4.4
Widowed	6	1.2
Educational level ^a		
No formal education	15	3.0
Primary school	192	38.7
Secondary school	219	44.2
Technical/vocational	44	8.9
Higher	26	5.2
Occupation		
Student	10	2.0
Housewife	364	73.4
Daily laborer	20	4.0
Private business	62	12.5
Employee	40	8.1
Estimated household monthly in in USD ($n = 411$) ^b	come	
< 56.7	109	26.5
≥56.7	302	73.5

 $^{\rm a}$ Educational level refers to the highest level of education attended, whether or not that level was completed

^b 85 missed values

[69/496 (13.9%)] newborns experienced complications within two hours after delivery (Table 4).

The magnitude of directly observed D&A among newborns Our data showed that all of the newborns [496/496 (100%)] experienced at least one form of D&A. The most frequently observed form of D&A was non-consented treatment for newborns who developed complications [69/69 (100%)], no mother was requested for consent before the provision of treatment for her sick newborn

 Table 3
 Obstetrics characteristics of mothers in selected public hospitals of Addis Ababa, Ethiopia, 2020 (n = 496)

Variables	Frequency	Percent
Place of birth		
Gandhi memorial hospital	284	57.3
Tikur Anbassa specialized hospital	142	28.6
Ras Desta memorial hospital	70	14.1
Parity ^a		
Nulliparous	312	62.9
Primiparous	113	22.8
Multiparous	58	11.7
Grand multiparous	13	2.6
ANC visit		
Yes	491	99.0
No	5	1.0
No. of ANC visits $(n = 491)$		
< 4	109	22.2
≥4	382	77.8
Referral status		
Self-referred	19	3.8
Referred from another facility	477	96.2
Mode of delivery		
Spontaneous vaginal delivery	279	56.3
Instrument-assisted vaginal delivery	134	27.0
Cesarean section	83	16.7
Maternal complication ^b		
Yes	75	15.1
No	421	84.9
Length of labor at the facility		
≤12 h	482	97.2
>12 h	14	2.8
HIV status		
Positive	4	0.8
Negative	492	99.2

^a Parity excludes the current childbirth

^b Maternal complications that occurred during labor, childbirth, or within the first two hours after delivery were a genital tear, postpartum hemorrhage, antepartum hemorrhage, severe preeclampsia, eclampsia, and anemia

baby. The next commonly observed forms of D&A were giving no information or explanation to parents or caregivers before or during the provision of ENC [464/496 (93.6%)], and lack of breastfeeding counseling and thermal support during the immediate postpartum period [433/496 (87.3%)].

Our data also showed that a large proportion of newborns were not given the recommended ENC: [210/496 (42.3%)] were not dried immediately, [306/496 (61.7%)] had their cord clamped within one minute after birth, [45/496 (9.1%)] were not placed on their mother's **Table 4** Health-related characteristics of newborns in selectedpublic hospitals of Addis Ababa, Ethiopia, 2020 (n = 496)

Variables	Frequency	Percent
Gestational age ^a		
28–36 weeks	84	16.9
37-41 weeks	341	68.8
≥42 weeks	71	14.3
Sex		
Female	259	52.2
Male	237	47.8
Birth weight in gram		
< 2000	5	1.0
2000–2500	89	17.9
>2500	402	81.1
Congenital malformation ^b		
Yes	3	0.6
No	493	99.4
Complication ^c		
Yes	69	13.9
No	427	86.1
HIV status ^d		
Exposed	4	0.8
Non-exposed	492	99.2

^a Gestational age is presented in completed weeks

^b Hydrocephalus, meningomyelocele, and meningocele

^c Complication that occurred within the first two hours after delivery included asphyxia, respiratory distress syndrome, and meconium aspiration syndrome

 $^{\rm d}$ HIV status of the newborn represents; exposed and non-exposed if born from HIV positive and HIV negative mother respectively

abdomen skin-to-skin immediately after birth, and [173/496 (34.9%)] were not breastfed within one hour after birth. In addition, [120/496 (24.2%)] and [9 (1.8%)] of the newborns were not given Vitamin K and Tetracycline eye ointment, respectively. It is recommended that healthy mothers and newborns should receive care in the facility for at least 24 h, however, our study showed [296/496(59.7%)] of newborns discharged before 24 h. Moreover, [122/496(24.6%)] of newborns shared a bed and [51/496(10.3%)] of mothers of the newborns reported neglected or their care delayed.

Similarly, a significant proportion of the newborns [204/496(41.2%)] experienced at least one form of physical abuse in which 115/496 (23.2%) of newborns with clear amniotic fluid who started breathing on their own received routine airway suctioning and a third of newborns [166/496 (33.5%)] were slapped on their back or held upside down.

The magnitude of each item of D&A is shown in Table 5.

Table 5 Directly observed D&A among newborns in selected public hospitals of Addis Ababa, Ethiopia, 2020 (n = 496)

Categories of Disrespect and abuse	Yes n (%)	No n (%)
Physical abuse	204 (41.1)	292 (58.9)
Airway suctioning without medical indication	115 (23.2)	381 (76.8)
Shaking or slapping or holding the newborn upside down	166 (33.5)	330 (66.5)
Stigma and discrimination		
Considering some ill babies "too sick to save" ^a	9 (13.0)	60 (87.0)
Discrimination against HIV exposed newborns ^b	3 (75.0)	1 (25.0)
Discrimination against newborns with congenital malformation ^c	2 (66.7)	1 (33.3)
Failure to meet professional standards of care	494 (99.6)	2 (0.4)
The newborn is not dried immediately after birth and is not well covered with a clean and dry towel	210 (42.3)	286 (57.7)
Newborn not placed on mother's abdomen skin-to-skin	45 (9.1)	451 (90.1)
Cord cut within one minute	306 (61.7)	190 (38.3)
Vitamin K was not administered	120 (24.2)	376 (75.8)
Tetracycline eye ointment was not administered	9 (1.8)	487 (98.2)
Breastfeeding is not initiated within the first hour after birth	173 (34.9)	323 (65.1)
The newborn is separated from the mother/parent/caregiver without medical indication	0 (0.0)	496 (100.0)
Non-consented treatment of a newborn (consent from parents or caregiver) ^a	69 (100.0)	0 (0.0)
Lack of privacy or confidentiality especially related to the HIV status of infant private (talking about it in a way that others can hear) ^b	1 (25.0)	3 (75.0)
Crowded conditions, shared beds	122 (24.6)	374 (75.4)
Early discharge from the facility (before 24 h)	296 (59.7)	200 (40.3)
No explanation is given to parents/caregivers at least once about the healthcare services provided to newborn	464 (93.6)	32 (6.4)
Neglect or delay in the provision of care while the newborn needed care at any point during a hospital stay	51 (10.3)	445 (89.7)
Poor rapport between patients and providers	433 (87.3)	63 (12.7)
Parent/caregiver is not counseled on breastfeeding counseling and thermal support during the immediate postpartum period	433 (87.3)	63 (12.7)

^a The denominator is newborns who experienced complication/s after delivery

^b The denominator is newborns who were born from HIV-positive mothers

^c The denominator is newborns who had congenital malformation/s

Factors associated with the observed D&A of newborns

Of the total newborns observed, six in ten [298/496 (60.1%)] of them 'received substandard care' the outcome variable used to measure D&A among newborns.

The multivariable logistic regression analysis showed that the health-related variables of the newborn (gestational age at delivery and newborn's sex), sociodemographic variables of the mother (place of residence and marital status), and obstetrics variables of the mother (number of ANC visits and mode of delivery) were found to have a statistically significant association with the outcome variable.

The odds of experiencing D&A among newborn was about two times more likely among preterm newborn (AOR=2.02, 95%CI: (1.11–3.69)) compared to term newborns, among female newborns (AOR=2.01, (1.37–2.95)) compared to male newborns, among newborns who were born from mothers who reside in rural areas (AOR=1.97, 95%CI: 1.22–3.20) compared to those who reside in urban areas, among newborns delivered through instrument-assisted delivery (AOR=2.19, (1.20–3.99)

compared to those who were delivered through Cesarean section; 2.7 times more likely among newborns of currently unmarried mothers (AOR=2.77, 95% CI:1.26– 6.06) compared to those who were married; and 2.4 times more likely among newborns whose mothers received ANC visits fewer than four times (AOR=2.37, 95% CI 1.42–3.96) compared to those who received four or more times (Table 6).

Discussion

Our study assessed the level of and factors associated with D&A among newborns through direct observation of childbirth care in three hospitals in Addis Ababa. We found that all newborns experienced at least one form of D&A. The proportion of newborns who experienced specific forms of D&A ranged from 0 to 100%. Gestational age at birth, sex of the newborn, maternal residence, maternal marital status, number of ANC visits, and mode of delivery were significantly associated with experiencing D&A among newborns.

Table 6	Factors associated with	observed D&A of newbor	ns in selected public ho	spitals of Addis Ababa, 2020
---------	-------------------------	------------------------	--------------------------	------------------------------

Variables	Substandard care		COR	AOR	
	No	Yes	95% Cl	95% CI	
Gestational age					
28–36 weeks	18 (21.4)	66 (78.6)	2.77(1.58-4.88)	2.02 (1.11-3.69)*	
37–41 weeks (Ref)	147 (43.1)	194 (56.9)	1	1	
≥42 weeks	33 (46.5)	38 (53.5)	0.87(0.52-1.46)	0.92(0.54-1.58)	
Sex					
Female	85 (32.8)	174 (67.2)	1.87(1.29-2.68)	2.01 (1.37-2.95)***	
Male (Ref)	113 (47.7)	124 (52.3)	1	1	
Residence					
Urban (Ref)	164 (43.2)	216 (56.8)	1	1	
Rural	34 (29.3)	82 (70.7)	1.83(1.16-2.87)	1.97(1.22-3.20)**	
Marital status					
Married (Ref)	189 (42.9)	251 (57.1)	1	1	
Unmarried, divorced, widowed	9 (16.1)	47 (83.9)	3.93(1.88-8.22)	2.77(1.26-6.06)*	
No. of ANC					
<4	25 (22.9)	84 (77.1)	2.72(1.67-4.44)	2.37(1.42-3.96)**	
≥4 (Ref)	171 (44.8)	211 (55.2)	1	1	
Mode of delivery					
Spontaneous vaginal delivery	114 (40.9)	165 (59.1)	1.48(0.91-2.43)	1.37(0.81-2.31)	
Instrument-assisted vaginal delivery	42 (31.3)	92 (68.7)	2.24(1.27-3.95)	2.19(1.20-3.99)**	
Cesarean section (Ref)	42 (50.6)	41 (49.4)	1	1	

Ref: Reference group

**p-value < 0.01

***<0.001

Previous studies reported a wide range of magnitude of D&A among newborns. In a study conducted in Nepal, none of the mothers reported their newborns experiencing any form of D&A during exit interviews [25]. However, there is a tendency to underreport negative experiences during exit interviews within the health facility compound. Studies conducted in Tanzania showed higher reports of D&A during household surveys and using direct observation compared with mothers' reports documented through exit interviews [30, 31].

Our study used a checklist gleaned from the first comprehensive systematic review that defined D&A among newborns [13], WHO's ENC guideline, and safe childbirth checklist [28, 29] and the updated respectful maternity care charter [2] to measure the various forms of D&A newborns might experience. The lack of consensus on what constitutes D&A among newborns and a standard measurement tool and data collection method makes the need to develop a standard validated measurement tool that comprehensively captures the different aspects of D&A among newborns more urgent.

The WHO recommends the provision of ENC immediately after birth including immediate drying, early skin-to-skin initiation, delayed cord care, early breastfeeding initiation, vitamin K and Tetracycline administration [29]. However, our study revealed a gap in the provision of these ENC. This is consistent with previous studies that reported a range of gaps in the provision of ENC components [14, 15, 21, 22]. Given Ethiopia's ambitious plan to meet the SDG targets, our finding indicates the need for a concerted effort by all stakeholders to increase the coverage of quality ENC services to ensure all newborns benefit from the care.

In our study, potentially harmful newborn care practices such as routine airway suctioning and shaking or slapping, or holding the newborn upside down were observed among one-fourth and one-third of the newborns, respectively. Compared to our data a recent multi-country facility-based observational study reported a higher proportion of routine airway suctioning (68%) and a relatively lower proportion of slapping or upside-down holding of the neonates (19%) [15]. These practices were commonly practiced in previous times. However, currently, the WHO recommends against them because of inconclusive evidence regarding their benefits or their negative outcomes [32, 33]. Further studies exploring

^{*}p-value < 0.05

the reasons why care providers continue to practice the potentially harmful practices and designing effective interventions to reduce them are needed.

Even though newborns are not able to communicate, their parents or caregivers are entitled to receive information and provide informed consent for their child's care in the child's best interest [2]. Our study found that all of the mothers whose newborns needed treatment were not asked for consent [69/69 (100%)] before treatment was given to their newborns. This was high compared to studies from Kenya (60%) and Nepal (63%) [14, 20]. The very high rate of non-consented care observed in our study could be partly because it assessed whether consent has been sought before the provision of any type of care for newborns with complications while the study from Kenya assessed consent before procedures and examination and the Nepal study before medical interventions. In addition, our study showed that the majority of mothers [464/496 (94%)] didn't receive any information or explanation about the routine care provided to their newborns. We argue that beyond the immediate unfavorable experience of mothers, these experiences might affect future healthcare-seeking and newborn care practices among mothers [20]. There is an urgent need for improving the communication between MNH providers and mothers or caretakers.

Our study also captured forms of D&A among newborns derived from health system deficiencies including crowded conditions/shared beds and the practice of discharging of newborns before 24 h after birth. These were linked to lack of space and a shortage of beds in the delivery rooms at the study hospitals. In addition, our data showed that one in ten newborns was neglected or received delayed care. This could be due to the high caseload at the study hospitals that overwhelm the MNH providers as a qualitative study finding from Kenya documented [34]. The health system should create an enabling environment for the health workers to provide quality care that ensures the safety and dignity of both the women and their newborns [2, 35].

In low-income countries, the smallest and sickest newborns particularly those born before 28 weeks of gestation, usually die [36] or are not at all considered viable. To improve their survival and well-being preterm newborns need extra care and attention during childbirth care and the postnatal period. Our data showed that preterm newborns were about two times more likely to experience D&A compared to term newborns. Given the pre-existing vulnerability of preterm newborns, failure to provide standard care would further reduce their chance of survival.

Our finding showed newborns whose mothers reside in rural areas and are currently unmarried were more likely to experience D&A compared to those living in urban areas and married. A significant association between a mother's socio-demographic characteristics and the mistreatment of newborns was observed in a previous study in Nepal [14]. In addition to conducting further studies to explore reasons for mistreatment based on socio-demographic characteristics, interventions aimed at improving healthcare provider's attitudes and acknowledging implicit biases need to be considered [37].

Our data also found that newborns whose mothers received fewer than four ANC visits were more likely to experience D&A compared to newborns whose mothers received four and more visits. Similarly, newborns delivered through instruments were more likely to experience D&A compared to those delivered through Caesarean section (C/S). Further studies exploring the reasons behind these are needed.

Our study has strengths and limitations. We believe the use of direct observation of childbirth and newborn care in our study provided us with a more objective assessment of D&A experiences among newborns. However, the direct observation might have also increased the likelihood of the healthcare providers behaving differently due to the observation. To minimize this, we did not show the specific checklist we used to assess the D&A and, it is also unlikely that they maintain their changed behavior during the relatively long observation period and throughout the data collection period. Although we intentionally included hospitals with different caseloads, our study was conducted at three facilities located in Addis Ababa limiting the representativeness of the data regarding newborn care provision at hospitals and health centers in the country. We also didn't limit the number of observations per provider; one healthcare provider was observed more than once. Thus, we weren't able to control the effect of health professional-related factors in our analysis. In addition, the convenience sampling technique that we employed limits the representativeness. For logistical reasons our observation duration lasted only until two hours after birth and, hence, our data did not capture the care provided afterward. Furthermore, due to the small sample size, we were not able to do a sub-analysis of the D&A experience for newborns with neonatal congenital malformation and who were HIV exposed. Finally, our observation didn't categorize D&A assessment by the severity of complications some of the newborns experienced. This might inflate the D&A prevalence among these groups of the infant population as some of the recommended routine and/or additional care may be withheld or delayed to them due to urgent interventions they need. Thus, we recommend further research overcoming these limitations.

Conclusion

Our study found a high magnitude D&A among newborns. The forms of D&A among newborns included the provision of substandard care and poor communication between mothers of newborns and the health care providers. Forms of D&A among newborns derived from health system deficiencies were also observed. Gestational age at birth, sex of the newborn, maternal residence, maternal marital status, number of ANC visits, and mode of delivery were significantly associated with the D&A. Interventions directed at the health system and individual level are recommended to improve the provision of respectful and dignified care to newborns.

Abbreviations

ANC	Antenatal care
AOR	Adjusted odds ratio
COR	Crude odds ratio
CI	Confidence interval
D&A	Disrespect and abuse
ENC	Essential newborn care
ETB	Ethiopian Birr
GMH	Gandhi Memorial Hospital
HIV	Human immunodeficiency virus
HMIS	Health Management and Information System
MNH	Maternal and newborn health
RDMH	Ras Desta memorial Hospital
SD	Standard deviation
SDG	Sustainable Development Goals
TASH	Tikur Anbessa Specialized Hospital
WHO	World Health Organization

Acknowledgements

We would like to express our gratitude to the School of Public Health, College of Health Sciences, Addis Ababa University for the financial support. We also would like to thank the Obstetrics and Gynaecology department heads of the study hospitals for providing us the opportunity to conduct the study and HMIS personnel for providing us with the necessary data. Furthermore, we gratefully acknowledge Dr. Aschalew Worku for his insights and contribution throughout the development of the study protocol and tools. Last, but not the least, we would like to thank the study participants for their voluntary participation and the data collectors for collecting quality data in a challenging environment.

Author contributions

ASE conceived the study. ASE and RG designed the study. RG coordinated the data collection and analyzed the data. ASE supervised the data analysis. RG wrote the original draft, ASE reviewed and substantively edited the draft. Both authors read and approved the final manuscript.

Funding

The author received no specific funding for this work.

Availability of data and materials

The datasets used and analyzed during the current study are available from the corresponding author upon reasonable request.

Declarations

Ethics approval and consent to participate

We obtained ethical approval from the Research Ethics Committee of the School of Public Health, College of Health Sciences, Addis Ababa University. In addition, we obtained permission to conduct the study from the three

hospitals. Before starting the data collection, the data collectors read to the mothers the information sheet containing details of the study objectives and procedures and provided copies of them to the MNH care providers for reading. After giving responses and clarifying to the study participants the data collectors sought verbal consent from mothers and written consent from health care providers.

Consent for publication

Not applicable.

Competing interests

The authors declare no competing interests.

Author details

¹Department of Reproductive, Family and Population Health, School of Public Health, Addis Ababa University, Addis Ababa, Ethiopia.

Received: 27 February 2023 Accepted: 18 August 2023 Published online: 31 August 2023

References

- World Health Organization. Standards for improving the quality of maternal and newborn care in health facilities. 2016.
- 2. White Ribbon Alliance. Respectful maternity care charter: universal rights of mothers and newborns. Washington. 2019.
- Paulson KR, Kamath AM, Alam T, Bienhoff K, Abady GG, et al. Global, regional, and national progress towards sustainable development goal 3.2 for neonatal and child health: all-cause and cause-specific mortality findings from the global burden of Disease Study 2019. Lancet. 2021;398:870–905.
- Central statistical agency (CSA)[Ethiopia] and ICF. Ethiopia Demographic and Health Survey, Addis Ababa, Ethiopia and Rockville. Maryland, USA: CSA and ICF; 2016.
- Ethiopian Public Health Institute (EPHI) [Ethiopia] and ICF. Ethiopia Mini demographic and Health Survey 2019: final report Rockville. Maryland, USA: EPHI and ICF; 2021.
- Bhutta ZA, Das JK, Bahl R, Lawn JE, Salam RA, et al. Can available interventions end preventable deaths in mothers, newborn babies, and stillbirths, and at what cost? Lancet. 2014;384:347–70.
- Kruk ME, Gage AD, Arsenault C, Jordan K, Leslie HH, et al. High-quality health systems in the Sustainable Development Goals era: time for a revolution. Lancet Glob Health. 2018;6:e1196-252.
- Gabrysch S, Nesbitt RC, Schoeps A, Hurt L, Soremekun S, et al. Does facility birth reduce maternal and perinatal mortality in Brong Ahafo, Ghana? A secondary analysis using data on 119 244 pregnancies from two cluster-randomised controlled trials. Lancet Glob Health. 2019;7:e1074-87.
- Bishanga DR, Massenga J, Mwanamsangu AH, Kim Y-M, George J, et al. Women's experience of facility-based childbirth care and receipt of an early postnatal check for herself and her newborn in Northwestern Tanzania. Int J Environ Res Public Health. 2019;16:481.
- Bruce SG, Blanchard AK, Gurav K, Roy A, Jayanna K, et al. Preferences for infant delivery site among pregnant women and new mothers in Northern Karnataka, India. BMC Pregnancy Childbirth. 2015;15:1–10.
- 11. Peca E, Sandberg J. Modeling the relationship between women's perceptions and future intention to use institutional maternity care in the Western Highlands of Guatemala. Reprod Health. 2018;15:1–17.
- 12. Sacks E, Kinney MV. Respectful maternal and newborn care: building a common agenda. Reprod Health. 2015;12:1–4.
- Sacks E. Defining disrespect and abuse of newborns: a review of the evidence and an expanded typology of respectful maternity care. Reprod Health. 2017;14:1–8.
- KC A, Moinuddin M, Kinney M, Sacks E, Gurung R, et al. Mistreatment of newborns after childbirth in health facilities in Nepal: results from a prospective cohort observational study. PLoS ONE. 2021;16:e0246352.
- Sacks E, Mehrtash H, Bohren M, Balde MD, Vogel JP, et al. The first 2 h after birth: prevalence and factors associated with neonatal care practices

from a multicountry, facility-based, observational study. Lancet Glob Health. 2021;9:e72–80.

- Shikuku DN, Milimo B, Ayebare E, Gisore P, Nalwadda G. Practice and outcomes of neonatal resuscitation for newborns with birth asphyxia at Kakamega County General Hospital, Kenya: a direct observation study. BMC Pediatr. 2018;18:1–11.
- 17. Bowser D, Hill K. Exploring evidence for disrespect and abuse in facilitybased childbirth. Boston: USAID-TRAction Project; 2010. p. 3.
- Waiswa P, Nyanzi S, Namusoko-Kalungi S, Peterson S, Tomson G, et al. I never thought that this baby would survive; I thought that it would die any time⁴: perceptions and care for preterm babies in eastern Uganda. Tropical Med Int Health. 2010;15:1140–7.
- Turan JM, Nyblade L. HIV-related stigma as a barrier to achievement of global PMTCT and maternal health goals: a review of the evidence. AIDS Behav. 2013;17:2528–39.
- 20. Nakphong MK, Sacks E, Opot J, Sudhinaraset M. Association between newborn separation, maternal consent and health outcomes: findings from a longitudinal survey in Kenya. BMJ Open. 2021;11:e045907.
- de Graft-Johnson J, Vesel L, Rosen HE, Rawlins B, Abwao S, et al. Cross-sectional observational assessment of quality of newborn care immediately after birth in health facilities across six sub-saharan african countries. BMJ Open. 2017;7:e014680.
- Fisseha G, Berhane Y, Worku A, Terefe W. Quality of the delivery services in health facilities in Northern Ethiopia. BMC Health Serv Res. 2017;17:1–7.
- McMahon SA, Mohan D, LeFevre AE, Mosha I, Mpembeni R, et al. You should go so that others can come; the role of facilities in determining an early departure after childbirth in Morogoro Region, Tanzania. BMC Pregnancy Childbirth. 2015;15:1–9.
- Bohren MA, Vogel JP, Hunter EC, Lutsiv O, Makh SK, et al. The mistreatment of women during childbirth in health facilities globally: a mixedmethods systematic review. PLoS Med. 2015;12:e1001847.
- Gurung R, Ruysen H, Sunny AK, Day LT, Penn-Kekana L, et al. Respectful maternal and newborn care: measurement in one EN-BIRTH study hospital in Nepal. BMC Pregnancy Childbirth. 2021;21:1–13.
- 26. United nation. Transforming our world: the 2030 agenda for sustainable development. New York: United Nation; 2016.
- Wassihun B, Deribe L, Worede N, Gultie T. Prevalence of disrespect and abuse of women during child birth and associated factors in Bahir Dar town, Ethiopia. Epidemiol Health. 2018;40:e2018029
- World Health Organization. WHO safe childbirth checklist implementation guide: improving the quality of facility-based delivery for mothers and newborns. 2015.
- World Health Organization. WHO recommendations on newborn health: guidelines approved by the WHO Guidelines Review Committee. World Health Organization; 2017.
- Freedman LP, Kujawski SA, Mbuyita S, Kuwawenaruwa A, Kruk ME, et al. Eye of the beholder? Observation versus self-report in the measurement of disrespect and abuse during facility-based childbirth. Reprod Health Matters. 2018;26:107–22.
- Kruk ME, Kujawski S, Mbaruku G, Ramsey K, Moyo W, et al. Disrespectful and abusive treatment during facility delivery in Tanzania: a facility and community survey. Health Policy Plann. 2018;33:e26–e33.
- 32. Neumann I, Mounsey AL, Das N. PURLs: Suctioning neonates at birth: Time to change our approach. J Fam Pract. 2014;63(8):461–2.
- World Health Organization. Basic newborn resuscitation: a practical guide. World Health Organization; 1998.
- Abuya T, Warren CE, Ndwiga C, Okondo C, Sacks E, et al. Manifestations, responses, and consequences of mistreatment of sick newborns and young infants and their parents in health facilities in Kenya. PLoS ONE. 2022;17:e0262637.
- Midwives'voices W. Midwives' realities: findings from a global consultation on providing quality midwifery care. Geneva: WHO; 2016.
- World Health Organization. Survive and thrive: transforming care for every small and sick newborn. 2019.
- Abuya T, Ndwiga C, Ritter J, Kanya L, Bellows B, et al. The effect of a multicomponent intervention on disrespect and abuse during childbirth in Kenya. BMC Pregnancy Childbirth. 2015;15:1–14.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

